Accepted Manuscript

Probabilistic normed load monitoring in large scale distributed systems using mobile agents

Moazam Ali, Susmit Bagchi

PII: DOI: Reference:	S0167-739X(18)30984-1 https://doi.org/10.1016/j.future.2019.01.053 FUTURE 4746
To appear in:	Future Generation Computer Systems
Received date :	25 April 2018
Revised date :	18 December 2018
Accepted date :	27 January 2019



Please cite this article as: M. Ali and S. Bagchi, Probabilistic normed load monitoring in large scale distributed systems using mobile agents, *Future Generation Computer Systems* (2019), https://doi.org/10.1016/j.future.2019.01.053

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Probabilistic Normed Load Monitoring in Large Scale Distributed Systems using Mobile Agents

Moazam Ali, Susmit Bagchi Department of Aerospace and Software Engineering (Informatics), Gyeongsang Na. nal University, Jinju, South Korea 660701 Corresponding author: Susmit Bagchi, profsbagchi@gm a.c. m

Abstract

Due to rapid advancements and developments in computing platform. there is a tremendous growth in heterogeneous distributed systems involving mobile device. In distributed systems, achieving better performance and efficient utilization of shat d resources are dependent on appropriate load monitoring mechanisms. In large-scale d stributed systems, performing load monitoring is a challenging task and, it effects in increas. There is a compared to a monitoring system intended to large-scale distributed mobile agent based load monitoring system intended to large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based design is an attractive approach for load monitoring in large-scale distributed mobile agent based load monitoring system is presented. The design is b and on a probabilistic normed estimation model and corresponding monitoring algorithms. The experimental evaluations and scalability analysis presented in this paper illustrate the behavior of agents and performance of the system under different load conditions. Moreover, a detailed qualitative as well as quantitative analysis of various mobile agent models are presented.

Keywords: Distributed Systems, Mo'ale Agents, Load Monitoring, Resource Utilization, Cloud Computing.

1. INTRODUCTION

A distributed system is the combination of independent nodes operating in a network and appears to the users as a single coherent system [6]. Distributed systems mean distributed processing in a shared environment to many nize computing time and to increase the overall performance. However, achieving bett, p rformance and reduction of computing time are subject to proper load balancing and lor d moniforing mechanisms [7, 31]. It means that, the overall processing load should be equally distable d mong all the nodes with the proper load monitoring mechanism to avoid unce the distributed system may not offer improved performance [34]. However, in gene to subserve that load balancing is an important factor in heterogeneous large-scale diamontary distributed systems are used to monitor available computing nodes in a system [25], without load monitoring, it is difficult to employ load balancing approaches to distributed systems. In large-scale distributed systems, load monitoring by a system administrator is a very difficult task [7, 26]. Moreover, load monitor creates frequent intercommunication in distributed systems while collecting node information repeatedly [8].

Furthermore, checking the current status of the large set of available nodes in a distributed systems is difficult in the presence of network latency, because getting the entrement status of all the available nodes incurs immense internetwork communications increasing response time and affecting system performance. Therefore, a mechanism is required to overcome these problems in determining the current status of the available nodes. Mobile agents are conserving technologies which has the ability to autonomously manage, design, implement and maintain distributed systems [3, 28]. Mobile agents are autonomous software entities betwing the ability to migrate through the network from node to node [1, 24]. The basic charactoris is of mobile agents are autonomy in behavior, social interaction, reactive to its environmental changes and goal driven execution [2]. A mobile agent created in one node can transport its "core" and "state" to another node in the network, where it continues its execution [5, 29]. The function of "code" is to start execution and the "state" determines the actions of a mobile agent in the destination node. Thus, mobile agent based load monitoring is an attractive approach in large-scale distributed systems.

1.1 MOTIVATION

Load monitoring mechanisms are designed to monitor distributed processing load to improve system performance and to enhance resource utilization in distributed systems [27]. In order to maximize the efficiency of distributed systems and for better utilization of the resources load monitoring mechanisms should be able to conjuste i. e current status of nodes. Researchers have proposed different approaches to load monitoring in distributed systems [1, 4]. A Grid Monitoring Architecture (GMA) is designed for grid computing by Global Grid Forum Performance Working Group [9]. The key feature of GMA is to provide a low-level specification that supports required functionality and, en, bles interoperability. However, in general, the GMA model is computationally expensive [9]. Ir GMA, intercommunication between producer and consumer for a large set of tasks you'd enhance computational complexity to a large extent. The space complexity of the GMA hand i is high because of the information storage including a set of events. The Ganglia is a sce able and ributed monitoring system design for clusters and grids to achieve high-performance computing systems [10]. Ganglia is based on the hierarchical design to form a federation of clusters and it depends on a multicast based listen or announce protocol to monitor the state of clus' rs 's well as cluster nodes. In ganglia, there is no mechanism to store information of produces and consumers. Tabu search algorithm is another neighborhood based search method which is vied in distributed systems for monitoring applications [11]. However, a constant collection of untailed performance attribute values of a large number of nodes is difficult. On the contrary we argue that the mobile agent-based load monitoring approaches have several advantage, such as [11, 30], (a) Reduced network load, (b) Network delay resolve, (c) Dynamic adar .auon (d) Fault tolerance and, (e) Goal-driven behavior.

In this paper, $\sqrt{2}$ present the design, implementation, and evaluation of mobile agents based load monitoring in a distributed system environment. The process load estimation is computed by employing a joint probability model and norm function, which is computationally inexpensive. The detailed scalability analysis is presented by using experimental evaluations as well as by following regression analytical model. The distinctive features of the proposed approach are,

- Autonomous load monitoring by migrating mobile agents based on the probabilistic norm.
- Reduction of waiting time of a node as well as the network load in order to increase the overall system performance.
- Updating real-time load information to monitoring node for decision making based on time intervals.
- Adaptive decision making by mobile agents depending on value ying status of a node.

Rest of the paper is organized as follows. Section 2 presents related work. Section 3 illustrates software architectural design and estimation model. Section 4 presents monitoring algorithms. Section 5 describes the implementation environment. Section 6 represents experimental evaluations. Section 7 presents a comparative analysis (i the proposed algorithm to other contemporary designs. Lastly, section 8 concludes the paper.

2. Related Work

There are various approaches available for load monitoring in distributed systems. In general, load monitoring in large-scale distributed systems requires a considerable amount of computational resources [7]. The researcher. Lave proposed model-based methods and algorithms for minimization of time and en .v of computations for profiling data related to performance and energy of servers having mulicove processors [35]. However, the algorithmic model considers specific server configuration in evaluation and, does not cover heterogeneous nodes including clients. Iosup, A et al. have proposed a monitoring architecture for control of grids, which is known as Toytle [167 11, core issues addressed by Toytle are grid-awareness, scalability and communication stan, ords. Toytle architecture inherits Global Grid Forum Grid Monitoring Architecture (GGFGMA) guidelines [16]. This architecture consists of three layers such as, a Distributed core layer a Fier? chical connection layer, and Local monitor layer. This mechanism ensures that the higrarching connection yields a highly scalable approach, where the data can be collected from di te. int and large-scale distributed systems with some degree of fault tolerance. Pivot Tracing is monitoring framework for distributed systems that addresses two important limitations [17'. The first limitation is that, most of the monitoring system information is recorded a priori. The second limitation is that, the information is stored in a component or machine centric way //hic i makes it very difficult to correlate events that cross these boundaries. The Pivot tracing conil nes dynamic instrumentation with a relational operator e.g. happened before join which gives users run-time ability to define arbitrary metrics. The Pivot tracing approach is able o filte, select and group based on events meaningful at other parts of the systems. Dan Cunter et al. have proposed a very lightweight instrumentation system for dynamic monitoring o high p rformance distributed applications [18]. The system is used to collect and aggregate detailed monitoring information from distributed applications. The system consists of four main me it ring components such as, application instrumentation, activation service, monitoring e ent receiver and archive feeder. Moreover, to achieve high-performance none of the above components can cause the pipeline to block while processing the data. Their proposed approach is used to deal with performance problems such as, low throughput and high latency by

determining a detailed end to end instrumentation of all components including applications, operating systems, hosts and networks. InteMon is designed for monitoring and data mining in large clusters [19]. InteMon can monitor more than 100 nodes of a data center. Lis approach uses the Simple Network Management Protocol (SNMP) and, MySQI da abase is used for storing monitored data. The advantage of this approach is the ability to autor atically analyze the monitored data in real time and to alert users for any anomalies [19]. Mobile agents are an efficient approach, which migrates from node to node in a connected network for gathering status information. Mobile agents are autonomous which means that y user is no longer needed to allocate mobile agents to nodes for migration. The researchers ¹ ave proposed a Mobile Agent-Based Server Resource Monitoring System (MABSRMS) bas d on 1 lobile-C library [12]. In MABSRMS, mobile agents call a low-level function to monitor server resources effectively and efficiently. The monitoring algorithm can be deployed on som indes or all monitoring nodes quickly, easily and silently. In MABSRMS there is no backup inechanism specified for the servers. The intercommunication between the nodes and the orver would enhance when a large number of nodes are interacting with a single so ver for load monitoring. Distributed Architecture for Monitoring and Diagnosis (DIAMOND) is a hierarchical and distributed cooperating agent model designed for distributed monitoring and diagnosis system [13]. The hierarchical model gives the advantage to predict the behavior of the system providing high performance and flexibility. The component Viag, 223 and monitoring (CDM) systems are modular to guarantee flexibility to alter when need. (in case of expansion of any module without any new line of code [13]. Localhost Information Cervice Agent (LISA) is a lightweight dynamic service capable of providing application monitoring, configuring system parameter and optimizing distributed applications [14]. LISA framework is independent and it can be deployed on any node architecture or operating systems. The framework (core system) is based on modules responsible for managing and moditoring other modules. This approach is capable to continuously monitor itself and detect any potential problems and selects a possible solution to overcome the problem. Precede ce Based Monitoring (PBM) algorithm is proposed for load monitoring in cloud architecture us nar age resources based on time, event and precedence [15]. Reduced Penalty Class Algo thm (RFCA) is used for negotiation and service level agreement between the centralized node of child and the consumers. In their proposed model, if the user request is not fulfilled by a articular cloud, then the agent of the current cloud migrates the request of the users to a neighboring cloud.

3. Software Archite ture Design

3.1. M nitor ng Agent Model

The architectural node of the designed Mobile agent monitoring system is illustrated in Figure 1. The proposed architecture has three main modules such as, (a) Agent Monitoring Module (AMM), (b) Agent Decision Module (ADM) and, (c) Agent Migration Module (AMMO). The agent module is used to collect the current status of the available resources in distributed systems. In addition, the agent is programmed to wait for a randomly varying time intervals to generate different samples of the currently available resources at different time purpose of collecting different samples is to compare the status of resources at different time

intervals with randomness. The ADM computes these samples of data for making decisions. The decision can be at multiple levels. For example, to re-compute the resources ... the condition for threshold doesn't match, otherwise forwarding it to AMMO. The AMMO is used us migrate the agent to appropriate nodes in distributed systems if a migration decision is mode autonomously. The background processes or daemon processes are those programs that he periodic or non-terminable. Most of these processes are part of the operating systems. The 'background processes have significance in managing system performance. System resource an classified into different types and functions such as, CPU, video card, hard drive and, memory. In this paper, we are referring RAM and CPU as the main available system resource in a used for computational purposes.



Figure 1: Schematic diagram of Mobile Agent Model

3.2. Architectural Desigi.

The internals of architectural design of our proposed monitoring system based on mobile agents are shown in Figure 2. The agent onioning module is consisting of sub-modules for systems data collection, for computing idle and used resources (RAM and CPU), randomization of sampling and, storage of sampling data for the decision-making process. The agent decision module is consisting of sub-nodules for retrieving sampling data, categorization of sampling data by monitoring logic and decision making for agent migration. The agent migration module is consisting of sub-modules for collecting updated system loads, selection of the appropriate node based on specified critaria and migrating the agent to the next node for computing the current status of system recoarces.

- **Data collettion r iodule (DCM):** The function of the data collection module is to start the dat a extracuon from the currently available resources. The mobile agent will collect the current st tus of resources i.e. RAM and CPU. In the first phase, the DCM module calculate. the total size of RAM and CPU. In the second phase, the DCM module control of processes that are currently holding the resources. In the third phase, agents send this information to the resource compute module (RCM).
- **Resource compute module (RCM):** The resource compute module is used to compute the current status of the resources. The RCM module collects the status related

information from DCM in order to compute and convert it into a recentage value. Calculating percentage value helps to efficiently determine to sow exactly the percentage usage of RAM and CPU is varying in random time intervals. Later on, this information is sent to the data sampling module.

- Data sampling module (DSM): The data sampling module will tak. he first sample and will send it to the storage buffer and then waits for a random amount of time. Next, the sampling module will continue and take the second sample from ^DCM and, this process will continue until the specified amount of sampling is done. The purpose of using a randomized waiting time is to make sure that the sample collected by the DSM module are distributed over time with relative uniformity. To a roid effects of rapidly transiting processes in a system, the randomization of waiting time is complexed.
- Storage buffer module (SBM): The storage buffer is used to store all the sampling results which are collected by the sampling module. The sampling module stores the status information of both the RAM and CPU. The SPM module will store the samples and then wait for the random amount of time to continue sample collection.



rigure 2. Mobile Agent Design Architecture

- Data retrieval module (DRM): The stored data is retrieved from the storage buffer by the data setrieval module. The stored data contains status information of available resources i.e. CFJ status and RAM status, which indicate that how much CPU as well as RAM is loaded in a node. The DRM module will instantaneously retrieve the sampling data from SPM. The purpose of this module is to get the sampling data and to send it to the will M module.
- **Mon.' oring logic module (MLM):** The monitoring logic module computes the joint probability variation of processing load in a node based on sampled data. The minimum joint probability value is 0 and the maximum joint probability is 1. The joint probability

is used to calculate two instantaneous events that are probably occurring at the same time. Thus, we are calculating joint random variations of CPU load and R/. I load of a node. By taking the joint probability of these resources, it is easy to make a decision about the status of a particular node (i.e. heavily loaded or lightly loaded) brised on a suitable norm function. The monitoring logic is the key module of our mobile a_B t-based monitoring model, where monitoring logic will categorize the status of system. Load according to the embedded logic to achieve the desired goal.

- Monitoring decision module (MDM): The monitoring decision module is used to make a decision either to migrate the agent or to re-compute the status of resources. The monitoring decision module will make a decision based on the current status of CPU and RAM. For example, if the joint probability of the resource is a of a node is greater than a specified threshold value, then the MDM will cufor current probability is less than the migrating agent module (MAM). However, if the joint probability is less than the specified threshold value, then the MDM module with cond the agent to execute the RCM module for re-computing the status information.
- Node info module (NIM): The function of the orde info module is to check the current status of existing nodes in the distributed order in order to determine the online or offline nodes. The NIM module is very important in our mobile agent monitoring architecture to ensure the reliability and to increase the performance of the system by regularly updating its database if a notion is a ded or removed.
- **Computing next node module (CNN14).** The computing next node module is used to select the appropriate node based on the specified criteria. The CNNM module is a dynamic module because its criteria for selecting node can be easily changed according to the need of the system. The criteria for selecting an appropriate node can be categorized as, (a) select such a node which has high level of processing and computing resources, (b) select a node which has enough free available resources, (c) select nearest or adjacent node and, (d) a nede which have fast and less congested network link. In our proposed model, we select up adjacent node to decrease the computation time.
- **Migrate agent mody** (MAM): Migrate agent module is used to migrate the agent to the next node for computing the current status of the node. This module takes two inputs: one input from the migration decision module and a second input from computing next node module. The note that or action decision module will confirm the migration of agent, while the computing next node module will specify the node where the agent will migrate.

3.3. Lc .a Estimation Model

In order to ca' culate the joint probability variations of CPU load and RAM load to get combined load status of the no e n, the estimation model will compute the joint probability of CPU load $P_n(C)$ and PAM load $P_n(R)$ variations over time. Three samples are collected randomly within an estimation time window to reduce enlargement of memory consumption by statistical datasets. This set of samples is sufficient to formulate probabilistic normed estimation of dynamics of load variations. High frequency sampling in an interval is avoided because, repeated samplings are done at multiple intervals and, past sampled datasets are terporarily kept in history for decision-making purposes. It effectively means that, more than three samples are taken into consideration for making decisions. Hence, the computation of the discrete joint probability of resource load in a time window at the node n is given belov.

$$t \in Z^+, t < 4,$$

 $\wp_n(t) = (P_n(C)|_t).(P_n(R)|_t)$

The relative triangular distances between the three samples of discrete joint probabilities are computed as,

(1)

(2)

(3)

$$a, b \in Z^+,$$

 $d_n(a,b) = |\wp_n(a) - \wp_n(b)|$

The minimum value of joint probabilities of three different time intervals are computed as,

$$v_n = \min\{d_n(a,b): a, b < 4\}$$

The joint resource load of the corresponding noa is computed by employing a norm function as,

$$||v|| = |v_n + avg(d_n(a,b))|$$
 (4)

It is easy to verify that Eqn. (4) is a norm because, $\forall t \in Z^+$, $|A_t| \ge 0$, $\forall k \in \Re$: $|kA_t| = |k| \cdot |A_t|$ and, $|A_{t=a}| + |A_{t=b}| \ge |A_{t=a} + A_{t=b}|$ where, $A = v_n + avg(d_n(a, b))$ at a time instant t.

The norm value ||v|| of a node is sed a, an input parameter for decision making by load monitoring algorithm of the mobile as ents.

4. The Algorithm Design

The algorithm is designe i ba ed on the execution logic of mobile agents. We have employed a mobile agent to collect surpus information from nodes. The agent is capable to migrate autonomously based i n d cision logic. The agent sends response messages from target nodes to the monitoring node. The designing of the monitoring algorithm and decision algorithm are given below.

4.1. Mo. itoring Algorithm

The monitoring algorithm executes to compute the status of currently available resources of a node in a specified amount of time. The pseudocode representation of the monitoring algorithm is presented in figure 3. The monitoring algorithm computes the available resources such as, cpu_load and ram_load of a node. The existing cpu_load and ram_load are computed in terms of percentage (%) to accurately determine the freely available resources of a node. These values of cpu_load and ram_load are stored in an array for further processing. In the next step, the

algorithm accesses and retrieves the associated stored values in order to c_lculate the joint probability for cpu_load and ram_load of a node. The purpose of calculating ______int probability is to determine the random variations of cpu_load and ram_load in order to generate a combined load status. The algorithm is designed to run and compute the joint probroilin / for at least three different time instances to efficiently and effectively calculate the cpu_i____d and ram_load in random time intervals within an estimation time window.



Figure 3. Pseudo-code representatio. of monitoring algorithm

The algorithm calculates d_{12} , d_{13} and, d_{23}). The minimum value of joint probabilities is computed and is stored in variable v_{11} dim in ity, the algorithm calculates the final cpu_load and ram_load of a node by following the norm function. The computed final value v (i.e. norm value) is passed as an input parameter to molitoring decision algorithm.

4.2. Deci ion Algorithms

The function c, decision algorithm is to process the input data from the monitoring algorithm for decision making. The input data processing in the decision algorithm is done in three distinct zones for current system load categorization such as, zone_1, zone_2 and, zone_3. The

distribution of zones is based on joint probability values assigned to each zone. The zone boundaries are heuristically set as, *zone* $_1 \Rightarrow (v > 0 \land v \le 0.24)$, *zone* $_2 \Rightarrow (..., 0.24 \land v \le 0.48)$ and, *zone* $_3 \Rightarrow (v > 0.48 \land v \le 0.72)$. We have divided the decision algorithm into three parts based on the current status of zones for easy understanding and description.

4.2.1. Decision Algorithm for Zone_1

In our proposed decision algorithm there are two important variables formed $v_history$ and $v_current$. The $v_history$ stores last updated zone information to construct zone dynamics and $v_current$ stores current zone status. This will result in charging the status of $v_history$ to $v_current$ in the next execution instant. The pseudocode representation of a decision algorithm for zone_1 is presented in Figure 4.

Data types:				
String v_history, v_current;				
double cpu_load, ram_load;				
integer msg_history;				
Initialization:				
cpu_load =0, ram_load = 0, msg_history = 😳				
Procedure:				
//for zone 1:				
monitoring_decision(){				
$if(cpu_load > 0.85 \lor ram_load > 0.35)$				
migrate(Agent);				
$elseif(v > 0 \land v \le 0.24) $				
v_current = zone_1;				
$if(((v_history == zone_1) \cdot (v_history == null)) \land v_current == zone_1) $				
<pre>send_msg(monitoring_nr le,< lc `d_any,node_id >);</pre>				
$v_history = v_current;$				
migrate(Agent);}				
$if(v_history == zone \ 2 \land \ current == zone_1)$ {				
compute(cpu_load ran loo l);				
$if(cpu_load > r m_load)_1$				
send_msg(moni:oring_node, < load_light_cpu, node_id >);				
$v_history = v_c$ rent;				
migrate(Age .t);				
}else{				
<pre>send_msg monitoring_node, < load_light_ram, < node_id >);</pre>				
v_histor_' = r_cur_ent;				
migrate(Ay, rt) }				
$if(v'_{,stor}) = zone_3 \land v_{current} = zone_1)$				
<pre>send_msg(m nitoring_node,<load_light_cpu ram,node_id="">);</load_light_cpu></pre>				
$v_{histe} = current;$				
r.igrat (Agent);}}}				

Figure 4. Pseudocode representation of decision algorithm for zone 1

In zone_1, "irs' une *monitoring_decision* function checks the cpu_load and ram_load of a node. If any of individual load estimation is greater than 0.85, then it will result in migration of agent without sending any message to the monitoring node. The *elseif* condition checks the joint

probability value to identify the current zone status of a node and to inform the monitoring node for a particular load migration. The decision algorithm checks the status hist $y'(v_history)$ and, if the history is "null" or "zone_1" then the mobile agent will send a message to monitoring node indicating that this node is lightly loaded enabling to send any processir g load to it. Next, the mobile agent updates the history to zone_1 and will migrate to the not node. The second condition will check if the $v_history$ is in zone_2 or not. According, the algorithm will compute the cpu_load and ram_load once more. If the cpu_load is greater than ram_load, then the mobile agent will send a message to the monitoring node for the second end will update zone dynamics history and will migrate to the next node. The third condition is to check whether $v_history$ contains value zone_3 or uny other zone. If it is in zone_3, then it means that this node has previously highly loaded and is changed its status to lightly loaded. As a result, the mobile agent sends a n esse_c to the monitoring node for transferring the light processing load to the target node. Next, it '.pdates $v_history$ of the node and migrates to next node.

4.2.2. Decision Algorithm for Zone_2

In continuation of the first part, the second part in uccision algorithm identifies zone_2, which signifies a moderately loaded zone. The pseudocode representation of zone_2 of decision algorithm is represented in Figure 5.

```
//for zone 2:
elseif(v > 0.24 \land v \le 0.48){
v_current = zone_2;
if(v_history == zone_1 \land v_curren_ = 2... 2)
compute(cpu_load , ram_load);
if(cpu\_load > ram\_load){
v_history = v_current;
migrate(Agent); }
else{
send_msg( monitoring_r)de < load_light_ram, < node_id >);
v_history = v_current;
migrate(Agent); }}
if(((v_history=zon(_?) \lor (v_history==null) \land (v_current==zone_2))
if(msg_history \ge 3)
msg_history = 0;
compute(cpu_loa ., ra n_load);
if(cpu_load > r_{u} \neg l ad)
send_msg(moritori. node, < load_ram, node_id >);
v_history = v :urv ent;
migrate(Ag_ rt) f
else{
send_msg,monitor.ng_node, < load_cpu, node_id >);
v_{histor} = v_{cu} rent;
migrate(.`gent) }
}else<sup>s</sup>
send _msg(, `onitoring_node, < load_light_cpu ram, node_id >);
v_hi.tory = i _current;
msg_h, +or / + +;
r _____(Agent); }}
if v_1 istory == zone_3 \wedge v_current == zone_2
v_{hi} tory = v_{current};
migrale(Agent); }
```

Figure 5. Pseudocode representation of decision algorithm for zone 2

According to the computed joint probability value, it will assign zone_2 to $: v_c$. *rent* variable. The algorithm will check the zone history and current zone status to make fully er decision. In the next step, the agent will compute the cpu load and ram load status c. the node. In case the cpu load is greater than ram load, the mobile agent will send a message to up monitoring node to send a process with light cpu load to this particular node. The "30ru m will update the v_history variable and will migrate to the next node. Otherwise, v her the ram load is greater than cpu_load, mobile agent sends a message to the monitoring node usend a process with light ram load. The algorithm will update the v_history and will mi rate to the next node. Based on available condition and zone dynamics, the algorithm will cherk the message history (msg_history). The message history plays a crucial role to word sending of the same message infinitely to monitoring node. If the message history is grea or .nan a predefined value (which is set to three in this case), then the algorithm will compute up cp. 10ad and ram load of the node once again. If the cpu_load is high then the agent sends a m ssage to the monitoring node to transfer processes with high ram load, otherwise, the ngen, ner ds a message to transfer processes with cpu load. Next, the algorithm updates the v_{histor} , variable and initiates migration of agent. If the message history is less than a predefine value, then the agent will send a message to the monitoring node to transfer light cpu lord and ram load to a particular node. Once the message is sent, the v_history variable is upda χ^2 and the message variable is incremented. Lastly, the agent is migrated to the next no \sim If he v_history variable contains zone 3 and *v_current* variable contains zone 2, then the alg. ritum will update the *v_history* and will migrate the agent. The reason for not taking any action by the monitoring node is to avoid an already moderately loaded node from being getting a burst of processing load within a short time interval.

4.2.3. Decision Algorithm for Zon

The third part of the decision a'_{20} thr. describes the dynamics of zone_3, which signifies a highly loaded zone. The pseud 3-code . presentation of zone_3 of decision algorithm is presented in Figure 6.

```
//for zone 3:
elseif(v > 0.48 \land v \le 0.7)
v_current = zone/3;
if(v_history == zon : 1 \cdot v_current == zone_3)
wait(Random();
Compute_Node_Statu .();
monitoring_aecision();
ł
if(v_history - zone_2 \land v_current == zone_3)
v_{histor} = o_{urrent};
migrate(A_{\lambda});
if(((v_history == zone_3) \lor (v_history == null) \land (v_current == zone_3))
send_msg(monitoring_node, < stop_sending_load, node_id >);
v_history = v_current ;
migrate(Agent);
}}
```

According to the joint probability value, the algorithm will assign zone_3 to a $v_current$ variable. If in a node, the $v_history$ variable contains zone_1 and $v_current$ variable contains zone_3, then the algorithm will wait for a random amount of time to repeat the estimation. The h gic of using the random amount of time is to probabilistically avoid the running processes tear to completion to be included within the estimation of the load. When the waiting time here over, the algorithm will recall *Compute_Node_Status* function and *monitoring_decision* function to evaluate the current zone status. If the $v_history$ contains zone_2 and v_currer contains zone_3, then the algorithm will update the $v_history$ variable and will enforce agent mi_ration without sending a message to the monitoring node. The reason for not sending a me_sage to. transferring extra load is to avoid overloading scenario. Finally, if the $v_history$ varial le cont ins zone_3 or null value and, the $v_current$ variable contains zone_3, then the agent will cont a message to monitoring node to avoid transferring the load to an already highly overlow led node. After sending the message to monitoring node $v_history$ variable is updated and the a_ent is migrated to next node.

5. Implementation Environment

5.1.1. Deployment Model

We have implemented our software architecture on testbed comprised of five nodes having completely connected topology. We have designed one node as a monitoring node and, remaining four nodes are targets executing robbile agents. The inherent properties of a mobile agent are autonomousity, goal orientation and, unplugent decision making [20]. The advantage of using mobile agents in distributed system is to reduce the communication overhead by collecting the updated status of the available and resources. Our proposed deployment model for mobile agent monitoring system is illustrated in Figure 7.



Figure 7. Deployment model of mobile agent monitoring systems

The mobile agent will first check the availability of a node where it will migrate next; if the node is alive then the agent will migrate. Otherwise, it will check for next available node. The node

information is stored a priori into the node information database. The database 's updated when a node is added or removed from the system. The feature of our proposed agent conitoring system is to enable the mobile agent to efficiently collect the status information of a node and to send the information to the monitoring node. Another feature of proposed monitoring is the improved overall response time. In the traditional load monitoring approaches, the current load values are calculated for all the available nodes and later the aggregated information. is sent as a response message to the monitoring node. On the contrary, our proposed r.oo 1 calculates the current status of nodes and immediately sends the information to the monitoring node. This mechanism will reduce the response time improving the overall system performance.

5.1.2. Implementation Architecture

Our proposed architecture is implemented in heterogeneou opchaing systems environment and Java programming language is used to deploy a mobile agent framework (i.e. Java Development Framework (JADE)). JADE is selected because it is a Java-Cased platform providing a simple and, efficient environment. Agents are implemented as containers in JADE and, are distributed among all the nodes in the network. The first container launched by the platform is known as "Main Container" and all other containers launcied and so on. The main container represents the bootstrap point. The containers are connected by the Internal Message Transport Protocol (IMTP). The main container components are *container table* (CT), global agent descriptor table (GADT), local agent descriptor table (LADT), a_{δ} or management system (AMS) and, directory facilitator (DF), as illustrated in Figur C. The container-1 components are global agent descriptor table (GADT) and, local agent descriptor table (LADT), as illustrated in Figur 8.



Fig re 8. S. hematic diagram of static main-container and static container-1

Moreover, the CT n anages a registry of the object references and transport addresses of all available containers of the platform. GADT manages a registry of all available agents in the platform helding their current status and location. The LADT contains its local agent's address and also mail tains global agent descriptor table. Agent management system monitors the entire platform and, it provides access to the white pages of the platform as well as manages the life cycle of the agent. Every agent must register with AMS in order to obtain valid AID. The

directory facilitator (DF) is the agent that implements the yellow pages service to any agent which wants to register its services or search for other available services.

The configuration of our mobile agent framework development and ... time environment specification is illustrated in Table 1. Our proposed mobile agent nonitoring algorithm is developed using Java eclipse IDE on top of JADE agent platform for monitoring distributed system. In addition, to test our proposed monitoring algorithm, we have used two additional load generation (benchmark) software modules which are, (a) *CPU Stress* and, (b) *Heavy Load*. The purpose of using this software is to generate various categories of 10. ^d on our target nodes to monitor the variations in algorithmic behavior under different load conditions. In terms of network connectivity, one of the nodes is wirelessly connected, and the rest of the nodes are wired connected with the monitoring node. The wired connected at 100Mbps through switch and wireless network operates on 100Mbps at peak.

Nodes	Specification	.ime Environment		
		Operating 2,m	Software	
Node 1	Intel Celeron G1840 CPU 2.80 GHz, RAM: 4 GB, HDD: 128 GB, NIC: Wireless Adaptor	Windows 10		
Node 2	Intel Core i7-6700 CPU 3.40 GHz, RAM: 8 GB, HDD: 2 TB, NIC: Realtek PCIe Controller	Linux arnel 2.6 Fedora	Eclipse 4.6, JADE 4.5.0, JDK 1.8, CPU Stress and Heavy Load.	
Node 3	Intel Core i5 3.1GHz, RA' <i>A</i> : 3 GB, HDD: 500 GB, NIC: Realtek PCIe Controll <i>c</i>	W idows 10		
Node 4	Intel Core 2 Duo E8 ⁽²⁰⁾ CPU 3.00 GHz, R .M: 3 GB, HDD: 320 Сь, NIC: Realtek PCIe Controllei	Windows 7		
Monitoring Node	Intel Core i7 670 CPU 3.40 GHz, 1. ^V. 8 GB, HDD: 2T 3, NIC. Pealtek PCIe G', E F mily Controller	Windows 10		
Network	Ethernet. 100 ps LAN Wir less: 100 Mbps WAP, Sig tal s rength: 45% (aver re)			

Table 1. Platform specifications of runtime environmen. and system configuration

The zonal 'ec sion-making algorithms are implemented as separate functions within the agent body (code). The total lines of code (LOC) for each agent are equal to 182.

6. Experimental Evaluations

The monitoring algorithm computes the status of currently available node resources in a specified amount of time. In order to evaluate the behavior and performance of our implemented system, we have conducted several sets of experiments under different local conditions. We have employed four standard benchmarks namely, (1) Heavy load, (2) CPU stress (3) FFT-z and, (4) DGEMM. These benchmarks exert non-uniform load stress at node resultated in sampled profiles presented in Figure 9, Figure 10, Figure 31, Figure 33, Figure 35 and, Figure 37. The stress profile samples indicate that, CPU and RAM variations are uncorrelated and non-uniform in nature. We have evaluated the performance by using a set of parameters, which are zone dynamics, rule firing density, load dynamics, migration frequency under different load conditions, resource utilization and, decision logic executio for the time intervals. The rules are fired by the decision logic module of the mobile agent during system monitoring. The migration frequency represents the frequency of migration of agents between different nodes.



6.1. Set 1: Experimenting with CPU utilization in terms of Heavy Load and CPU Stress Benchmar's

In this case, the cpu_load is _mployed for estimating the behavior of zone dynamics, rule firing dynamics, the combined dynamics of CPU/RAM utilization, joint probability and, migration frequency of agents. 1. variation in zone dynamics with respect to increasing cpu_load is illustrated in Figure 11. The variations in rule firing density are illustrated in Figure 12. The variations in the combined dynamics of CPU/RAM and joint probability are illustrated in Figure 13. The variation frequency are illustrated in Figure 14.



Figure 11. Zone dynamics of mobile agent monitoring system (ex-set-1)





6.2. Set 2: Experimenting with CPU utilization in torus of Heavy Load and video on demand (VOD) Benchmarks

In this case, the cpu_load and video on demand (VOL) benchmarks are employed for estimating the behavior of zone dynamics, rule firing density, <u>combined</u> dynamics of CPU/RAM utilization, joint probability and migration frequency. The variation in zone dynamics with respect to increasing cpu_load and VOD is illustrated in Timere 15. Initially, cpu_load is in zone_1, however gradual increase in the load results in a 'ansition of the node to zone_3 and it stays there without migration. However, due to back ground processes, an abrupt increase in the load happens exceeding the specified threshold value. This results in agent migration. The variations in rule firing density are illustrated in Figure 16. The variations in the combined dynamics of CPU/RAM utilization and, joint probability are illustrated in Figure 17. Under increased cpu_load and VOD, the status of CPU utilization becomes maximum. The variations in migration frequency are illustrated in Timere 18. It is shown in the figure that increasing the CPU load and VOD causes frequent migration or agent between different zones.



Figure 15. Zone dynamics fine ile agent monitoring system (2x-set-2)





6.3. Set 3: Experimenting with CPU utilization is. terms of Heavy Load, CPU Stress and video on demand (VOD) Benchmarks

In this case, the cpu_load is employed by apply. To heavy load generation software, VOD and CPU stress software for estimating the behavior of zone dynamics, rule firing dynamics, combined dynamics of CPU/RAM, joint proteinity and, migration frequency. The variation in zone dynamics with respect to increasing cpu_load is illustrated in Figure 19. The variations in rule firing density are illustrated in Figure 20. The variations in the combined dynamics of CPU/RAM utilization and non-probability are illustrated in Figure 21. The variations in migration frequency are illustrated in Figure 22.



Figure 19. Zone dynamics of $n_{s}^{+1}e_{s}$ agent monitoring system (excert-3)



Figure 21. Combined dynamics of CPU and RAM load of mobile agent monitoring system (ex-set-3)



Figure 20. Rule firing density of mobile agent monitoring system (exset-3)



Figure 22. Migration frequency distribution of agent under cpu_load and VOD load (ex-set-3)

6.4. Set 4: Experimenting with CPU and RAM utilization in terms of Aea. ^y Load and CPU Stress Benchmarks

In this case, the cpu_load and ram_load are employed by applying heavy load generation software, VOD and CPU stress software for estimating the behavior of one dynamics, rule firing dynamics, combined dynamics of CPU/RAM utilization, join probability and, migration frequency of our proposed algorithmic approach. The variation in zero dynamics with respect to increasing cpu_load is illustrated in Figure 23. The variations in the firms density are illustrated in Figure 24. The variations in the combined dynamics of CPU RAM tilization and, computed joint probability are illustrated in Figure 25. The variations in mig. tilization and, requered and figure 26.



6.5. Set 5: Experimenting with RAM utilization in terms of Heavy Load and video on demand (VO)) Ben hmarks

In this case, the cpu load and ram_load is employed by applying heavy load generation software and VOD fc estimating the behavior of zone dynamics, rule firing dynamics, combined dynamics of CFUTAAM utilization, joint probability and, migration frequency. The variation in zone dynamics with respect to increasing cpu_load is illustrated in Figure 27. The variations in rule firing density are illustrated in Figure 28. The variations in the combined dynamics of CPU/RAM utilization and, joint probability are illustrated in Figure 29. The variations in migration frequency are illustrated in Figure 30.



Figure 29. Combined dynamics of CPU and RAM load of mobile agent monitoring system (ex-set-5)

Figure 30. Migration frequency distribution of agent under cpu_load and VOD load (ex-set-5)

6.6. FFT-z and DGEM[•] 1 Stress Test

In this section, the experiment 1 s tup and performance of the monitoring algorithm are described by using resour a utilization and decision logic execution. We have conducted four set of experiments (ex-1 to ex-1) under different load distribution in order to evaluate the decision logic execution senerated by monitoring agents. Moreover, we have evaluated the decision logic executio. Vy monitoring agent using a set of numerically tagged decision logic parameters such as, 1: any_load, 2: compute_cpu_ram_load, 3: light_cpu_load, 4: *light_ram_load*, 5. ¹*i t_pu_ram_load*, 6: *migrate_agent*, 7: *ram_load*, 8: *cpu_load* and, 9: stop_sending_1_ud. In the first experiment, the CPU and RAM utilization is evaluated using FFT-z CPU s ress benchmark having CPU cores saturation as core-C0, core-C2, core-C4 and, core-C6 normalizations in resource utilization and decision logic execution for this experime t are illustrated in Figure 31 and Figure 32 respectively. It is evident that the behavior of decision logic execution is periodic in nature. The reason for periodic execution behav. . . . due to the current load having average ranged values (CPU 50% and RAM 60%). In the second experiment, the saturation of CPU cores is set to core-C0, core-C1, core-C3, core-C4 and, core-C6 as illustrated in Figure 33. The behavior of decision logic execution is aperiodic in nature. The reason for aperiodicity is due to the algorithmic logic of monitoring

algorithm to execute various decision outputs as illustrated in Figure 34. In the third experiment, the saturation of CPU cores is set as core-C0, core-C1, core C2, core-C3, core-C4, core-C5, core-C6 and, core-C7 as illustrated in Figure 35. In this case, the a verage value of resource utilization of CPU is 100% and utilization of the RAM i, 60%. The reason for full utilization of CPU is due to saturating maximum available number of activated CPU cores. The decision logic execution of monitoring agent is periodic in nature due to full utilization of CPU exceeding the specified threshold value. There as illustrated in the Figure 36 the decision logic 9 (stop sending load) is validated. In the fourth experiment, the stress test is carried out by the DGEMM CPU stress benchrunk as l'lustrated in Figure 37. The DGEMM is a stress test model based on multithreaded program ming. The sets of matrix dimensions for DGEMM are consisting of 150x150, 500x560, 10 J0x1000 and 1500x1500. The average resource utilization of CPU is 40% while the ... M utilization is 70% on the average. The decision logic execution sequence of agent moniloring algorithm is illustrated in Figure 38. In this case, the mobile agent executes decision logic 5 (*light_cpu_ram_load*) and 8 (cpu_load) in most cases. The reason for fit yent execution of decision logic 5 and decision logic 8 is due to the current lightly nodeu status of a node. Therefore, the monitoring node can send CPU load as well as <u>Light Data</u> load to a particular node.



Figure 31. Resource utilization of me⁻¹e agent monitoring system (ex-1





Snap shot of decision logic execution of algorithm

Figure 33. Resource utilization of mobile agent monitoring system (ex-2)

Figure 34. Decision logic execution of mobile agent monitoring system (ex-2)



Figure 37. Resource utilization of mobile agent monitoring system Figure 38. Decision logic execution of mobile agent monitoring system (ex-4) system (ex-4)

6.7. Decision Execution 1. file / f Agents

In our monitoring algorithm, leci ion execution block of our mobile agent is designed to fire decision sequences according to the computing status of currently available resources at a node at an instant of time. To evaluate the decision sequences generated by agents, we have conducted five experiments (x-1 to ex-5) under different load conditions. We have evaluated the decisior secuences by using a set of numerically tagged decision logic given by, 1: any_load, 2: mpute_cpu_ram_load, 3: light_cpu_load, 4: light_ram_load, 5: light_cpu_ram_lor d, 6 migrate_agent, 7: ram_load, 8: cpu_load and, 9: stop_sending_load. The decision exection p of the represents the sequence of fired decision logic by the mobile agent in the sp chie time interval. In the first experiment, the dynamism of CPU utilization is evaluated in erms o Heavy load and CPU stress benchmarks. The variation in decision execution sequences for experiment one is illustrated in Figure 39 (the x-axis represents time and y-axi represent numerically tagged decision logic). The behavior of decision execution sequence h peri dic, because the current load status of the node is less than the threshold value. In the second experiment, the dynamism of CPU utilization is evaluated in terms of Heavy load and video on demand (VOD) benchmarks. The variation in decision execution sequences for the second experiment is illustrated in Figure 40. The behavior of decision logic sequence is aperiodic because of the gradual increase in the CPU load. In the third

experiment, the dynamism of CPU utilization is evaluated in terms of Heavy load, CPU stress and, VOD combined benchmarks. The variation in decision exection sequences for the third experiment is illustrated in Figure 41. In this case, the mobile agent initially fires decision logic 5 (light_cpu_ram_load) based on current load status. However, further increased in load results in firing decision logic 9 (stop_sending_load, because the load is now increased from the specified threshold value in the current node. In the fourth experiment, the dynamism of CPU and RAM utilizations are evaluated in terms of Heavy load and CPU stress benchmarks. The variation in decision we ution sequences for the fourth experiment with respect to increasing CPU and RAM Lad is .'lustrated in Figure 42. In the fifth experiment, the dynamism of RAM utilization is valuat d in terms of Heavy load and VOD benchmarks. The variation in decision execution sequences for the fifth experiment with respect to increasing RAM load is illustrated in Figure '... In this case, the mobile agent initially executes decision logic 1 (any_load), which means the the current load status of the node is marked as very lightly loaded and the monitorn 7 node can send any load to this particular node. After more than halftime of the conducting experiment, the mobile agent executes decision logic 2 (compute_cpu_ram_loaa, and o (cpu_load). However, the overall decision logic sequence remains stable at sequence 1 (uny_load) for a relatively long period of time, because of the lightly loaded status of the *i*, spective node.



Figure 39. Decision logic sequence of the agent monitoring system (ex-1



Figure 41. L •cision logic sequence of mobile agent monitoring system (ex-3)



Figure 40. Decision logic sequence of mobile agent monitoring system (ex-2)



Figure 42. Decision logic sequence of mobile agent monitoring system (ex-4)



Figure 43. Decision logic sequence of mobile agent monitoring system (ex-5)

6.8. Scalability estimation

Scalability is the ability of a system, a network or an application to maintain overall performance at an acceptable level even if the load is increased. In this section, we have conducted experiments to evaluate the scalability of the propored design architecture by computing response time of mobile agents. Our experiment is consisting of a group of nine computing nodes at maximum, which are connected by heterogeneous networks (wired and wireless networks). We have considered heterogeneous hardware platforms to prepare testbed, which are comprised of: (a) nodes having dual-core Celeron 2.8GHz CPU and limited RAM (4GB) and, (b) nodes having core if 3.4GHz corboard 8 CPUs) and 8GB RAM. We have measured response time as an average of the corresponded experiments on the testbed for each node counts. We have conducted scalability analysis in two phases such as, (1) experimentally measuring on the testbed and, (2) the correspondent projecting through regression analysis for a very large set of nodes based on our experimental data.



In Figure 44, experimentally measured response time is presented for wired network, wireless network and hybrid network (wired and wireless). It is illustrated in Figure 44 that, the average response time of nodes connected by a wired network is less than wireless connected nodes. This is because hydrogeneous available bandwidth of the wired network is greater than the wireless network. Moreover, the wired medium is less prone to data losses and crosstalk as compared to wireless

medium. In order to analyze the behavior of our proposed model for a very larg set of nodes, we have performed a regression analysis for projection based on Equation 5.

(5)

$$y_i = b_0 + b_1(x_i)$$

The values of coefficients b_0 and b_1 are computed by using the experimental data of Figure 44. The values of b_0 and b_1 are given in Table 2.





700000 600000 500000 (ms) 400000 300000 200000 100000 0 100 1000 1500 500 Number of Nodes wired — — wireless

Figure 45. Projection of response time of agent through regression analysis

The projection of res_{F} or set i me of mobile agents for a very large set of nodes is illustrated in Figure 45. The regression a alysis shows that, for a very large set of nodes, the response time of agents tends to diverge considering wired and wireless networks. The wired network connected nodes are having better response time due to high bandwidth and less interference in data communication.

7. Comparative Analysis

We have evaluated our mobile agent-based load monitoring model with various agent-based load monitoring frameworks for performance analysis. Some of these include, *Agent-Based Adaptive Monitoring System* [21], *Java Based Agent Management System* [23], *Localhost Information*

Service Agent (LISA) [14], Web Server Load Monitoring System using a Mol ile Agent [12], A Method of Network Monitoring by Mobile Agents [1], A Mobile Agent-Based System for Server Resource Monitoring [2] and, Monitoring Agents in A Large Integrated Services Architecture (MonALISA) [22]. Each mobile agent model is examined and evaluated with respect to various design parameters and attributes like autonomisity, reliability, mobil. network latency, heterogeneity and, flexibility. A detailed discussion is explained in the following sections.

7.1. Qualitative Analysis

In our qualitative analysis, we have selected six parameters $f_{\mathcal{O}1}$ qualitative analysis namely, autonomousity, reliability, mobility, network latency, flexibility and, heterogeneity as illustrated in Table 3. In the qualitative analysis, the parameters are $c^{\text{pt}}eg_{\mathcal{O}1}$ in between [0, 1] scale, where 0 indicates the low value and 1 indicates high value $M'_{\mathcal{O}1}ec$ ver, within 0 and 1 scale we have categorized the interval into three zones (first zone is from 0 to 0.3 which indicates low zone (L), the second zone is from 0.3 to 0.6 indicating mean zone (M) and, the last zone is from 0.6 to 1 indicating high zone (H)).

7.1.1. Autonomousity

In case of autonomousity, the agent-based adaptive in onitoring system (ABAMS) is in high zone, because the system administrator does not need to update the system information when new nodes are appended (the information up'rting s carried out by the mobile agent) [21]. The autonomousity of mobile agent based notwork monitoring and management system (MABNMMS) is in high zone, because the agene will migrate to various nodes in the network, gather status information and carry out device control tasks [1]. This model is designed for decentralized networks without auton usity. The autonomousity of Monitoring Agents in A Large Integrated Services Architect re (Mc nALISA) is in a high zone, because this model is designed as an autonomous, multi-u. and ded agent-based monitoring system. MonALISA registers dynamic services whic' ar ab'e to collaborate and cooperate for performing a wide range of monitoring tasks in dis. "br.ed systems [22]. Without autonomous behavior, the collaborating and cooperating, with other agents will be difficult to achieve. The autonomousity of Mobile Agent-Based Server ke ource Monitoring System (MABSRMS) is in high zone, because this is designed to collect available system resources in the large scale-distributed systems [12]. This model ses three different mobile agents to collect system resources. The autonomousity of Dis.rib ted Management System based on Java agents (DMSBA) is in the medium zone, becaus, the ranagement is handled by the administrator, but when the system loses contact with control auministrator or having problems on time scales, the mobile agent will take over and a tonom usly manages the system and performs automated tasks [23]. The autonomousity of Matile Agent-Based Load Monitoring (MABLM) is in the high zone because the agent is d signed in such a way that it autonomously and automatically gathers system status information and serus the results to monitoring server [20]. The autonomousity of Localhost Informatic a Carrie Agent (LISA) is in high zone, because this monitoring system is designed to dynamically onfigure system parameters and autonomously migrates agent to various nodes in the network to increase the system performance [14]. The autonomousity of our mobile agentbased monitoring system is in a high zone, because the mobile agent collects system status information automatically. In our model, the agent must be autonomous, becar se the agent will make a decision about migrating to the next node. If the node is not available doen the agent will select the next available node in the network.

7.1.2. Reliability

Reliability is used to measure the system stability. The reliability on vent-based adaptive monitoring system (ABAMS) is in the low zone, because server hok r is used for collecting information of active nodes by sending heartbeat signals [21]. The r liability of mobile agent based network monitoring and management system (MABNM 3) is in high zone, because in this model fault management module is used to monitor failts and take necessary actions accordingly to increase overall system reliability [1]. The reliability of the MonALISA monitoring system is in a high zone, because this model use a d in mic pool of threads [22]. The reliability of Mobile Agent-Based Server Resource Monitoring System (MABSRMS) is in the medium zone, because it uses more than one server for enhanding reliability [12]. The reliability of Distributed Management System based on Java agen. (DM; BA) is in high zone, because this model uses mobile agents that can be trained to perform a set of tasks such as, monitoring host network traffic and restarting a specific server the has been crashed recently to achieve a high level of reliability [23]. The reliability of Mobile Agen. Based Load Monitoring (MABLM) is in the medium zone, because this model uses master-state agents for monitoring [20]. The master agent keeps track of slave agents in the system. The reliability of Localhost Information Service Agent (LISA) is in high zone, because this than work uses the binary module, which can be automatically restarted by the mobile and an ating reliable monitoring system [14]. The reliability of our mobile agent-based monitoring system is in a high zone, because our mobile agent checks the status of each node. If some node goes offline then the agent will immediately notify the monitoring node and the r onitoring node will take necessary action accordingly.

7.1.3. Mobility

In our qualitative analysis, we 'lave tale ted seven models which are based on mobile agents. All these models are in the high tare in terms of mobility, because mobility is an essential factor of agents.

7.1.4. Network 1 atracy

In the case of network 'atency ABAMS model is in a high zone, because the agent will communicate with set 2° broker, not with monitoring system directly [21]. The communication between server broker and monitoring node will increase intercommunication, which in turn increases the net rork la ency. Network latency of MABNMMS monitoring system is in the medium zone, because the mobile agent collects status information of all available nodes in the network and ends is to the manager node [1]. The mobile agent does not send individual node status information which increases the delay in the monitoring process. Network latency of MonALIS and which increases the network load as well as network latency increasing overall system performance [22]. Network latency of MABSRMS model is in the medium zone, because the network latency of model uses three different agents for collecting system information, which increases the network

latency. Network latency of the DMSBA model is in the medium zone, because this model replicates data on multiple servers for reliability, which increases the net ork latency [23]. Network latency of MABLM model is in the medium zone, because the master agent will create slave agents and then it migrates slave agents to collect system information for available nodes in the network [20]. Network latency of LISA monitoring model is in the low one, because LISA monitoring system can dynamically load or unload its binary modules to promor current status and sends results directly to a monitoring system [14]. Moreover, ne work latency of our mobile agent-based monitoring system is in the low zone, because the provide agent collects status information and directly sends to the monitoring node.

7.1.5. Flexibility

In the case of flexibility, ABAMS model is in a high zone, because each monitoring component can be dynamically activated, deactivated, moved or changed vithout affecting or restarting the entire monitoring system [21]. The flexibility of MAPNM. ⁴C model is in the low zone, because the module or component cannot be changed once the system is deployed [1]. The flexibility of MonALISA monitoring model is in the medium .one, because this framework allows cooperating services [22]. The flexibility of MAFCPMS model is in the low zone. The reason is that, this model has no mechanism to adopt ch. "ses in its monitoring component [12]. The flexibility of the DMSBA model is in a high rone because this model has the capability to dynamically load new code to existing code fo, enhancing agent capabilities and functionality [23]. The flexibility of MABLM monitoring model is in the low zone, because once the model has deployed, there no mechanism to upgrade its existing components [20]. The flexibility of the LISA monitoring model is in a high 7 one, c cause LISA framework is flexible enough to include new modules to the existing system v. thout compromising the performance and efficiency [14]. The flexibility of our monitoring model is in the medium zone, because we can upgrade its monitoring functionality before deploy nent, but once it is deployed we cannot change its component.

7.1.6. Heterogeneitv

The heterogeneity of AI AMS model is in a high zone, because this model uses dynamically controllable agents in 1 distributed network to utilize heterogeneous resources efficiently. This model uses an agent which can be loaded, unloaded on a specific resource or it can be migrated to another node depending on the requirement. Heterogeneity of MABNMMS monitoring framework is in high one, because it can be deployed on various nodes having a different platform in a distributed system. Heterogeneity of MonALISA is in high zone, because this model allows cooperating services and adaptation to a dynamic environment. Moreover, cooperating services and adaptation to a dynamic environment. Moreover, cooperating services also make this monitoring framework very efficient when monitoring a large number of incerogeneous nodes having different response times. The heterogeneity of the DMSBA n. Ndc i is in a high zone, because the interoperability of Java allows this architecture to be highly ada table and, portable. The heterogeneity of MABLM monitoring model is in high zone, because this monitoring framework uses Mobile Agent Server (MAS), which provides an

environment for executing mobile agents on various nodes (heterogeneous nodes) in distributed systems. Heterogeneity of LISA framework heterogeneity is in high zone. The heterogeneity is managed by the ability to detect various architectures and deploying its buncry modules dynamically to monitor its current status. Heterogeneity of our mobile agent nonitoring system is in a high zone, because our monitoring model is developed using an a language, which provides interoperability and can be easily deployed in heterogeneous planarms.

Deremators	Autonomousity	Poliobility	Mahility	Natwork	- Tlovik (ity	Hataraganaitu
Faralleters	Autonomousity	Kenability	Mobility	INCLWOIK	I Maxin Inty	neterogeneity
				later .y		
Agent-						
Based						
Models			4			
ABAMS	Н	L	Н	h	Н	Н
MABNMMS	Н	Н	Н	IVI	L	Н
MonALISA	Н	Н	Н	т	М	Н
MABSRMS	Н	М	H	М	L	М
DMSBA	М	Н	Н	М	Н	Н
MABLM	Н	М	<u>h</u>	М	L	Н
LISA	Н	Н	H	L	Н	Н
MABMS	Н	Н	ЧН	L	М	Н

 Table 3. Comparative analysis of load monitoring models

Legends: H: High, M: Medium, L: Low, agent-based adaptive monitoring system (ABAMS), Mobile agent-based network monitoring and mail regement system (MABNMMS), Monitoring Agents in A Large Integrated Services Archi reture MonALISA), Mobile Agent-Based Server Resource Monitoring System (MABSRMS', Disulting ed Management System based on java agents (DMSBA), Mobile Agent-Based Load Mc nitoring (MABLM), Localhost Information Service Agent (LISA), Mobile Agent-Based Monitoring Cyster (MABMS).

7.2. Quantitati /e / nalysis

In this section, we will qualitatively analyze different agent-based monitoring models. The comparative studies *i* a *r* erformed based on fault tolerance, scalability, mobile agent migration latency and, response the *a* illustrated in Figure 46, 47, 48, 49 and, 50, respectively. The values of performance matrix a determined with approximation by analysis. The metric values are set in a scale betwee. 0 and 1, where 0 represents the lowest value and, 1 represents the highest value. We have divided the interval into three zones such as, the first zone is from 0 to 0.3 (low zone), the second zor *e* is from 0.3 to 0.6 (medium zone) and, the last zone is from 0.6 to 1 (high zone).

7.2.1. Scalability

Scalability is defined as the ability of a system or model that describes it, car ability to perform efficiently under an increasing number of nodes. The scalability of ABANY framework is in a high zone, because it can monitor more than one multicast groups, however, the administrator can add or remove multicast group without disturbing the system performance [21]. MABNMMS monitoring system has scalability in the medium zon, + ecause this framework is tested on a fixed number of nodes [1]. Scalability of MonALIS ... frame work is in the medium zone, because it is designed as an autonomous multi-threa led sei-describing agent-based system, which is able to collaborate and cooperate in performing rarious monitoring task by adding more nodes in the existing system without affecting syst an performance [22]. Moreover, the scalability of MABSRMS monitoring framework is in the medium zone, because it is deployed in large-scale distributed systems. The mobile a ent calls low-level functions in dynamic libraries and can monitor system resources efficiently vithout considering the increased number of nodes [12]. The scalability of the DMSBA monitoring system is in the medium zone. It is designed for large-scale distributed systems [22]. The scalability of MABLM is in high zone, because it is designed for large-scale distributed system and operates efficiently where a large number of systems are connected through . ne work having low bandwidth and high delay time [20]. The scalability of LISA framewor' is in high zone, because the agents automatically detect the architecture and load binary modules (in case of the appended node) necessary to perform monitoring services making this _____ rk scalable [14]. The scalability of our mobile agent monitoring system is in a high zone, uscause the mobile agent detects any addition or removal of nodes in the system and updates its information in the node information database.

7.2.2. Fault Tolerance

In case of fault tolerant, ABAM⁶, fre new ork is in the low zone, because there is no mechanism specified which can be used to recover from the failure [21]. For example, if the monitoring agents die or stop working d' v to some error, then the entire system fails. The fault tolerance of MABNMMS monitoring system is in a high zone, because the fault management module uses a mobile agent to monitor the retwork and identify any faults and take a necessary control action [1]. The fault tolerance of MonALISA framework is in high zone, because this framework assigns an independent thread to each task so that if some task fails due to error other tasks should not be affected which make this framework fault tolerant [22]. The fault tolerance of MABSRMS moniforing system is in the low zone, because there is no backup mechanism in monitoring syster [12]. ' he fault tolerance of DMSBA framework is in a high zone, because the agents detect failure. congestion automatically and, change the granularity at which the data is collected [23]. This nechanism provides the administrator with a detailed view of the system. The fault tolerance of MABLM monitoring system is in the medium zone. The agents communic us with each other if some fault or error occurs and agents notify the monitoring system whit, taking control action enhancing fault tolerance [20]. The fault tolerance of LISA monitoring system is in a high zone, because the core system manages monitoring modules and dynamically restarts the crashed module [14]. The fault tolerance of our mobile agent-based

monitoring system is in a high zone, because the mobile agent migrates to each node and collects status information. If any node dies or an error occurs, then the mobile "Cent immediately informs the monitoring system. The mobile agent constantly updates the node information database, which enhances overall system fault tolerance.



7.2.3. Mobile Agent Migration Latency

In Figures 48 and 49, the comparison of mobile coent migration latency (excluding agent code execution time at nodes) is illustrated for the wind and wireless nodes for varying data sizes of agents. As illustrated in Figure 48 and 49, the agent behavior in the wired node is relatively stable without major increase or decrease in latency, because the wired connection is more reliable and provides a stable connection to upport bandwidth for a large volume of data. While in Figure 48 and 49, the behavior of wireless node is aperiodic and tends to overshoot and undershoot with respect to network load and congestion of wireless network increasing the agent migration latency in the wireless for work. The decreased latency values at instances 1, 3, 5 and 9, which is almost close to the wired node that at particular instances due to less congestion (packet loss is almost none and s_{12} all strength is good). The mobile agent migration latency is not greatly affected by the dat volume it carries in the network.



Mobile Agent Migration and Data IO Latency

Figure 48. Comparison of mobile agent migration and data IO latency with respect to wired and wireless node (data size = 1kb)



7.2.4. Execution-response Time

In Figure 50, the comparison of execution-response time (i.e. averaged value of agent code execution time at nodes excluding network IO and migration time) for an 3-node model is illustrated considering MABNMMS and MABMS. As illustrated in Figure 50, the execution-response time of MABMS is less than MABNMMS, which shows that MAB. 'S performance is better than MABNMMS. The MABNMMS monitoring agents use a field script [1]. The results illustrate that, the execution-response time depends upon the complexit of the script as well as the number of available nodes in the network. Execution-response time increases with the increase in a number of nodes in the network and, vice versa. While MABMS uses Java-based agents, the Java-based agents are platform independent and lightweight in nature. This reduces the execution-response time of agents in MABMS based on MABMS based on MABMS.



Figure 50. Comparison of sponse time for 3-Node triangular model

8. Conclusion

In large-scale distributed systems, k eping track of load monitoring mechanism by a system administrator is a very difficult u. '. L/ad monitoring mechanism is used to monitor available computing nodes in a distributed system. In this paper, we present the design, implementation, and evaluation of mobile agents 'ased load monitoring system for distributed systems in a heterogeneous network er vir, nment. Moreover, the monitoring algorithm runs to compute the status of currently availading node resources in time intervals. We used mobile agents to collect status information from podes and sending a response message to the monitoring node. The pseudocodes describing ioni oring algorithms and decision algorithm are presented. In order to evaluate the behavier and performance of our implemented system, we have conducted five sets of experiments ut der dit rent load scenarios. We have evaluated the performance using a set of parameters such as, 70° dynamics, rule firing density, load dynamics and, migration frequency under different lo. 1 conditions. A detailed scalability analysis is performed based on experimental value on on testbed and through regression analysis. The proposed approach reduces the waiting time of a node as well as the network load in order to increase the overall system perimance. The waiting time of a node is reduced by autonomously sending the required information to the monitoring node for decision making at different time instants.

References

- [1] Manvi, S, S and Venkataram, p. A method of network monitoring by mobile agents, computing, 2000, 2(3), pp. 4-5.
- [2] Aridor., Yariv., and Danny B. L., Agent design patterns: elements of agent application design. In Proceedings of the second international conference on Autonomous agents, ACM, 1998, pp. 108-115.
- [3] Das, Shantanu., Mobile agents in distributed computing: Networ c exploration. Bulletin of EATCS 1, no. 109, 2013.
- [4] Park, H. J., K. J. Jyung., and S. S. Kim., Mobile age." oase I load monitoring system for the safety web server environment. In International Conference on Computational Science, Springer, 2004, pp. 274-280.
- [5] Mostafa., Salama A., Mohd S. A., Muthukkarup, an A., Azhana A., and Saraswathy S. G., A dynamically adjustable autonomic again mamework. In Advances in Information Systems and Technologies, Springer, 2013, pp 621-642.
- [6] Xu F., Liu F., Liu L., Jin H., and Li L iAware: Making live migration of virtual machines interference-aware in the clove, TELE Transactions on Computers, volume: 63, IEEE, 2014, pp. 3012-3025.
- [7] Rajani S., and Garg N. A clustered app.oach for load balancing in distributed systems, International Journal of Mobile Computing & Application, volume: 2, SSRG-IJMCA, 2015, ISSN: 2393-9141.
- [8] Haverkamp DS., and Gav h S. Intelligent information agents: review and challenges for distributed information so. r.es, Journal of the Association for Information Science and Technology 49, no. 4, 998, page(s) 304-311.
- [9] Wang, Xiaoguang, ¹¹ui Wang, and Yongbin Wang. A unified monitoring framework for distributed enviro mer t. Intelligent Information Management 2, no. 07, 2010, pp. 398.
- [10] Massie, Matth w L., Drent N. Chun, and David E. Culler. The ganglia distributed monitoring system: d sign, implementation, and experience. Parallel Computing 30, no. 7, 2004, pp 217-8-32.
- [11] Vidhate, Sonali L, and M. U. Kharat. Resource Aware Monitoring in Distributed System using Taou Search Algorithm. International Journal of Computer Applications 96, no. 23 2014.
- [12] Ti, **7** A Mobile Agent-Based System for Server Resource Monitoring. Cybernetics and Info. r ation Technologies, 13(4), 2013, pp. 104-117.

- [13] Wörn, H., Längle, T., Albert, M., Kazi, A., Brighenti, A., Seijo, S.R., Senior, C., Bobi, M.A.S. and Collado, J.V., Diamond: distributed multi-agent architec'. A for monitoring and diagnosis. Production planning & control, 2004, 15(2), pp. 189-200.
- [14] Dobre, C., Voicu, R., Muraru, A. and Legrand, I.C., A distribute. agent based system to control and coordinate large scale data transfers. arXiv preprint a. Xiv: 196.5171, 2011.
- [15] Seenuvasan, P., Kannan, A. and Varalakshmi, P., Agent-Bas d keepurce Management In A Cloud Environment. Appl. Math, 11(3), 2017, pp. 777-786.
- [16] Iosup, A., Țăpuş, N. and Vialle, S., 2005, February. A monitoring architecture for control grids. In European Grid Conference, Springer, 2005, pp. >22-93..
- [17] Mace, J., Roelke, R. and Fonseca, R., Pivot tracing: Dynamic causal monitoring for distributed systems. In Proceedings of the 25th Symposium on Operating Systems Principles, ACM, 2015, pp. 378-393.
- [18] Gunter, D., Tierney, B., Jackson, K., Lee, J. and Chaffer, M., Dynamic monitoring of high-performance distributed applications. In High Performance Distributed Computing, 11th IEEE International Symposium, IEEE, 2002, pp. 163-170.
- [19] Hoke, E., Sun, J., and Faloutsos, C., In emon. Intelligent system monitoring on large clusters. In Proceedings of the 32nd international conference on Very large data bases, VLDB Endowment, ACM, 2006, pp. 12,39-1242.
- [20] Kim, S.T., Park, H.J. and Kim, Y.C., The load monitoring of Web server using mobile agent. In Info-tech and Info-net, 2001. Proceedings. ICII 2001-Beijing. 2001 International Conferences on IEEE, Vol. 5, pp. 89-94.
- [21] Kwon, S. and Choi, J., An age based adaptive monitoring system. In Pacific Rim International Workshop on Multi-Agents, Springer, Berlin, Heidelberg, 2006, pp. 672-677.
- [22] Legrand, I., Newman, H., Voicu, R., Cirstoiu, C., Grigoras, C., Dobre, C., Muraru, A., Costan, A., Dediu M, and Stratan, C., MonALISA: An agent based, dynamic service system to monter. control and optimize distributed systems. *Computer Physics Communicatior s*, 180(12), 2009, pp.2472-2498.
- [23] Brooks, C., L. nev, B. and Johnston, W., JAVA agents for distributed system management. *LBNL Report*, 1997.
- [24] Adacal, M. and Lener, A.B., Mobile web services: A new agent-based framework. *IEEE Intern & Conputing*, 10(3), 2006, pp.58-65.
- [25] Tomarching J., Vita, L. and Puliafito, A., Active monitoring in grid environments using mobile agant technology. In Active Middleware Services, Springer, Boston, MA, 2000, pp. 5, 66.

- [26] Du, T.C., Li, E.Y. and Chang, A.P., Mobile agents in distributed network management. *Communications of the ACM*, 46(7), 2003, pp.127-132.
- [27] Ahn, J., Fault-tolerant Mobile Agent-based Monitoring Mechanism. R⁺ Highly Dynamic Distributed Networks. *IJCSI International Journal of Compute* Science Issues, 7(3), 2010, pp.1-7.
- [28] Papavassiliou, S., Puliafito, A., Tomarchio, O. and Ye, J., Mobi engent-based approach for efficient network management and resource adjustation: framework and applications. *IEEE Journal on Selected Areas in Communications*, 20(4), 2002, pp.858-872.
- [29] Ku, H., Luderer, G.W. and Subbiah, B., Nover over An intelligent mobile agent framework for distributed network manageme. In *Global Telecommunications Conference, GLOBECOM'97.,1997, IEEE*, Vol. 1, p. 160-164.
- [30] Tomarchio, O. and Vita, L., On the use of mob. cod technology for monitoring Grid system. In *Cluster Computing and the Grid, Proceedings. First IEEE/ACM International Symposium on*, IEEE, 2001, pp. 450-455.
- [31] Iranpour E., Sharifian S., A distributed C. 1 balancing and admission control algorithm based on fuzzy type-2 and game theory for large-scale SaaS cloud architectures, Future Generation Computer Systems, Vol. 80 Tisevier, 2018, pp. 81-98.
- [32] Volkova, V.N., Chemenkaya, L.V. Den atirikova, E.N., Hajali, M., Khodar, A. and Osama, A., Load balancing in cloud computing. In *IEEE Conference of Russian Young Researchers in Electrical and Electronic Engineering (EIConRus), IEEE*, 2018, pp. 387-390.
- [33] Dasoriya, R., Kotadiya, P., Arya, G., Nayak, P. and Mistry, K., Dynamic load balancing in cloud a data-centric approach. In *Networks & Advances in Computational Technologies (NetACT - IEEE, 2017, pp. 162-166.*)
- [34] Lastovetsky, A., Szustak, L., & Wyrzykowski, R., Model-based optimization of EULAG kernel on Intel X on hi through load imbalancing. IEEE Transactions on Parallel and Distributed Systems, EEE, 2017, 28(3), 787-797.
- [35] Lastovetsky, Y., & Manumachu, R. R., New model-based methods and algorithms for performance and Y ergy optimization of data parallel applications on homogeneous multicore cluster. IEEE Transactions on Parallel and Distributed Systems, IEEE, 2017, 28(4), 1115–1132.

Moazam Ali obtained his BCS in Computer Science in year 2007 from the Islamia Collect. University of Peshawar and, MS-IT in Computer Networks in year 2012 from the Institute of Contagement Sciences, Peshawar. Currently, he is pursuing his PhD in Distributed Systems in the Department of Aerospace and Software Engineering (Informatics), Gyeongsang National University, Jinju, South Kolman.

Susmit Bagchi has received B.Sc. (Honours) from Calcutta University II. 1993, B. E. in Electronics Engineering in 1997 from Nagpur University, M.E. in Electronics and relecommunication Engineering in 1999 from Bengal Engineering and Science University (present relEST). He has obtained Ph.D. (Engineering) in Information Technology in 2008 from Bengal Engineering and Science University. Currently, he is Associate Professor in Department of Aerospace and soft vare Engineering (Informatics), Gyeongsang National University, Jinju, South Korea. His reseal th interdusts are in Distributed Computing and Systems.

ACCEPTED MANUSCRIPT



Moazam Ali



Susmit Bagchi

- Distributed systems load monitoring using agents and probabilistic r orm.
- Adaptive and autonomous load monitoring in distributed systems raing agents.
- Mobile agent and probabilistic norm based autonomous load monitoring ... distributed systems.